Author's response to reviews

Title: Induction of autoantibodies against lung matrix proteins and smoke-induced inflammation in mice

Authors:

Corry-Anke Brandsma (c.a.brandsma@path.umcg.nl)
Wim Timens (w.timens@path.umcg.nl)
Marie Geerlings (m.geerlings@path.umcg.nl)
Henrike Jekel (h.jekel@path.umcg.nl)
Dirkje S Postma (d.s.postma@long.umcg.nl)
Machteld N Hylkema (m.n.hylkema@path.umcg.nl)
Huib AM Kerstjens (h.a.m.kerstjens@long.umcg.nl)

Version: 4 Date: 10 December 2010

Author's response to reviews:

Dear editors of BMC Pulmonary Medicine,

I checked my manuscript for typographic errors and changed the items you requested. Here is a point-by-point description of the changes made;

• Affiliations are written in full according to the requested format.
• I have updated the references and corrected the order in which they are cited in the text.
• I have renamed Introduction into Background and Material and Methods into Methods.
• I removed all highlighting from the manuscript.
• I inserted an extra heading titled Conclusions at the end of the discussion.
• I changed the format of the table as requested.
• I changed the figures and removed the figure numbering from the figures.

Best wishes
Dr. Corry-Anke Brandsma